Author's response to reviews

Title: A Randomized Controlled Trial of an Extensive Lifestyle Management Intervention (ELMI) Following Cardiac Rehabilitation: Study Design and Baseline Data

Authors:

Scott A Lear (slear@providencehealth.bc.ca)
Andrew Ignaszewski (aignaszewski@providencehealth.bc.ca)
Wolfgang Linden (wlinden@vanhosp.bc.ca)
Anka Brozic (abrozic@vanhosp.bc.ca)
Marla Kiess (mkiess@providencehealth.bc.ca)
John J Spinelli (jspinelli@bccancer.bc.ca)
P. Haydn Pritchard (haydn@unixg.ubc.ca)
Jiri J Frohlich (jifr@unixg.ubc.ca)

Version: 3 Date: 24 Sep 2002

Revised manuscript based on copy edits.

Please note, Figure 2 is now missing a greater than/equals symbol in the Lipid Management box. This goes between HDL-C and 0.9 mmol/L in the middle row of the Lipid Management box.